Author’s response to reviews

Title: A simple and short microbiology practical improves undergraduate nursing students' awareness of bacterial traits and ability to avoid spreading infections

Authors:

RIKA YANO (r-yano@med.hokudai.ac.jp)
Torahiko Okubo (t.okubo@hs.hokudai.ac.jp)
Tomoko Shimoda (tshimoda@hs.hokudai.ac.jp)
Junji Matsuo (matsuo@hs.hokudai.ac.jp)
Hiroyuki Yamaguchi (hiroyuki@med.hokudai.ac.jp)

Version: 1 Date: 03 Dec 2017

Author’s response to reviews:

BMC Medical Education,
Dr. Sigrid Harendza, Editor

December 3, 2017

Thank you for your e-mail of 29 November, 2017 and returning our manuscript "A simple and short microbiology practical improves undergraduate nursing students' awareness of bacterial traits and ability to avoid spreading infections" (MEED-D-17-00780), to BMC Medical Education.

I am submitting our revised version for publication. According to the editor’s comments and suggestions, we revised our manuscript properly.

We look forward to hearing from the Journal as to the acceptability of this manuscript.

Thank you in advance for your kind consideration of this paper.
Sincerely yours,

Rika Yano
Hokkaido University
Kita 12, Nishi 5, Kita-ku, Sapporo 060-0812 Japan
Phone number +81-11-706-3323
Fax number +81-11-706-4911
r-yano@med.hokudai.ac.jp